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Another clear and cold morning today, with temperatures again in the low teens. The rest of the day was very similar to yesterday, with beautiful clear 
skies and a high around freezing. Winds were mostly calm, but increased in the afternoon. The weekend looks to be a bit warmer and sunny, before the 
next weather system moves in around Tuesday.

As is customary at this point, we had to reset HRXL’s at CC, SP, and DCS first thing in the morning. The morning data check showed no other new issues, 
but the battery at DC had dropped to near 12.5V, necessitating a battery swap. While we were there, Liz helped troubleshoot the electrical issues with the 
electrician and I took a soil sample and measured the boom angle with the Leica GPS antenna. The ground was quite frozen and covered in about 6 
inches of snow. For the electrical, it turns out there was a short at a junction outside of the barn. The electrician planned on bypassing that section as a 
temporary fix, so we will hopefully have power at the site restored by the time of this blog post. 

In the afternoon we returned to DCS to troubleshoot the thermocouple at 32m. Unfortunately the problem was with the thermocouple itself, forcing us to 
lower the tower to replace it. This did the trick and we are now getting data. We only have one spare thermocouple left at that site, but there is discussion 
about removing all the thermocouples from the towers before next week's forecast snow, which would alleviate this concern. We also decided to dress the 
cables at the tower to facilitate lowering and raising the tower. Thanks to Liz for doing much of the untangling and taping despite freezing hands! This 
should make life much easier for future visits to DCS. 

We wrapped up the day with the 4:15 pm sounding and a data check, finding no issues. Tomorrow is a down day for the project. 
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